Author's response to reviews

Title: Assessing medically unexplained symptoms: Evaluation of a shortened version of the SOMS-2 for use in primary care

Authors:

Cristina Fabião (cristinafabiao@gmail.com)
MC Silva (carolina@icbas.up.pt)
António Barbosa (abarbosa@netcabo.pt)
Manuela Fleming (manuelafleming@gmail.com)
Winfried Rief (rief@staff.uni-marburg.de)

Version: 8 Date: 29 April 2010

Author's response to reviews: see over
COVER LETTER

1- We update title
2- Corrected references
3- Wrote reference 32 in the text